

Citation: Tadege B, Mekonnen Z, Dana D, Sharew B, Dereje E, Loha E, et al. (2022) Assessment of environmental contamination with soil-transmitted helminths life stages at school compounds, households and open markets in Jimma Town, Ethiopia. PLoS Negl Trop Dis 16(4): e0010307. https://doi.org/10.1371/journal.pntd.0010307

Editor: Michael Templeton, Imperial College London, UNITED KINGDOM

Received: October 27, 2021

Accepted: March 7, 2022

Published: April 4, 2022

Copyright: © 2022 Tadege et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting information files.

Funding: The study was supported by a VLIR-UOS Network University Network (https://www.vliruos. be/en). BT received a scholarship within the same VLIR-UOS Network University Network (https:// www.vliruos.be/en). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Assessment of environmental contamination with soil-transmitted helminths life stages at school compounds, households and open markets in Jimma Town, Ethiopia

Bamlaku Tadege^{1,2,3}*, Zeleke Mekonnen^{2,4}, Daniel Dana^{2,4}, Bizuwarek Sharew², Eden Dereje⁴, Eskindir Loha^{5,6}, Jaco J. Verweij⁷, Stijn Casaert³, Johnny Vlaminck³, Mio Ayana^{2,3,4}, Bruno Levecke³*

 School of Medicine, Hawassa University, Hawassa, Ethiopia, 2 School of Medical Laboratory Sciences, Jimma University, Jimma, Ethiopia, 3 Department of Translational Physiology, Infectiology and Public Health, Ghent University, Merelbeke, Belgium, 4 Molecular Biology and NTDs Research Center, Jimma University, Jimma, Ethiopia, 5 Chr. Michelsen Institute, Bergen, Norway, 6 Centre for International Health, University of Bergen, Bergen, Norway, 7 Microvida, Laboratory for Medical Microbiology and Immunology, Elisabeth-TweeSteden Hospital, Tilburg, The Netherlands

* abelbamlaku@yahoo.com (BT); bruno.levecke@ugent.be (BL)

Abstract

Background

It remains largely unknown where and how infections with soil-transmitted helminths (STHs; *Ascaris, Trichuris, Necator* and *Ancylostoma*) occur. We therefore aimed to identify possible sources of infection by assessing the environmental contamination in an STH-endemic area.

Methods

We first performed a series of laboratory experiments designed to optimize a soil strainingflotation method to detect and quantify *Ascaris* and *Trichuris* eggs in soil, and to validate the diagnostic performance of the optimized method when followed by microscopy and qPCR. In a second phase, we applied this method to assess the level of STH contamination in 399 environmental samples collected from 10 school compounds, 50 households and 9 open markets in Jimma Town (Ethiopia). Subsequently, we explored associations between the environmental contamination and both the corresponding STH epidemiology at the level of the schools and the household characteristics. Finally, we assessed the knowledge, attitude and practice (KAP) towards STHs in school children.

Principal findings

Our soil straining-flotation method has an analytical sensitivity of 50 eggs per 100 grams of soil and egg recovery rate of 36.0% (*Ascaris*) and 8.0% (*Trichuris*). The analysis of field samples with both microscopy and qPCR revealed the presence of 8 different helminth species of medical importance, including but not limited to the human STHs. There was a

Competing interests: The authors have declared that no competing interests exist.

significant association between the environmental contamination and prevalence of any STH infections at the school level only. The KAP indicated a lack of knowledge and awareness of STHs.

Conclusions/Significance

Our optimized straining-flotation method has a moderate diagnostic performance and revealed that life stages of helminths are ubiquitous in the environment, which might be due to the poor sanitary facilities at both the schools and the households, and a poor level of KAP towards STHs. Further research is required to gain more insights into the contribution of these life stages to transmission.

Author summary

The recent strategies to control the morbidity caused by intestinal worms have mainly been focusing on large scale deworming programs during which drugs are administered to at-risk populations. Although these large-scale deworming programs have been successful in reducing the disease burden, re-infection in the absence of other intervention measures is unavoidable. Although there is a consensus on how infections with intestinal worms are transmitted, it remains unclear where hot spots of infectious life stages can be found in endemic communities, which in turn impedes both the design and the validation of interventions that prevent (re-)infection. We optimized a laboratory method to detect life stages in soil samples and applied this method to examine soil contamination at school compounds, households and open markets in Jimma Town (Ethiopia). Finally, we assessed the household characteristics, and the knowledge, attitude, and practice towards intestinal worms among school children. Generally, our results indicated (i) a moderate performance of our laboratory method, (ii) the ubiquitous presence of life stages in the environment, (iii) a poor level of sanitary facilities at both the schools and the households and (iv) a poor level of knowledge, attitude and practices towards intestinal worms.

Introduction

Soil-transmitted helminths (STHs) are a diverse group of intestinal worms, including the giant roundworm (*Ascaris lumbricoides*), the whipworm (*Trichuris trichiura*) and the hookworm (*Necator americanus, Ancylostoma duodenale* and *Ancylostoma ceylanicum*) [1]. Despite the clear biological differences across these worm species [2]; their transmission is characterized by the same sequence of events: (i) infected individuals excrete worm eggs via their stool into the soil; (ii) under optimal conditions of moisture and temperature the excreted eggs will develop into infectious worm eggs (*A. lumbricoides* and *T. trichiura*) /larvae (hookworms); (iii) finally infection occurs through oral uptake (eggs) or skin penetration (larvae) of these infectious stages that reside in the soil [3].

Globally, it is estimated that over one fifth of the world population are infected with at least one STH species, resulting in an estimated total disease burden of 3.4 million disabilityadjusted life years (DALYs) [4,5]. However, transmission of STHs and the associated disease burden mainly occurs in sub-tropical and tropical countries, where the climate is optimal for development of the infectious stages [6] and where the challenge to maintain an adequate personal hygiene and a lack of sanitation facilitate the transmission. STH-attributable morbidity is mainly associated with moderate-to-heavy intensity infections and disproportionately affects both children and women of child-bearing age [7,8], contributing to impaired growth and cognitive development, malnutrition, anemia, and school absenteeism [9–12]. To fight the STH-attributable disease burden, the World Health Organization (WHO) promotes largescale deworming programs in which anthelmintic drugs (albendazole and mebendazole) are periodically administered to at-risk populations (e.g., preschool-age children, school-age children, women of childbearing age and workers at higher risk for STH infections, such as teapickers and miners), with the ultimate ambition to reduce the prevalence of moderate-toheavy intensity infections to less than 2% by 2030 [13].

Although these campaigns have been successful in reducing the global STH-attributable morbidity [14,15], re-infection perpetuates in absence of improved water, sanitation and both environmental and personal hygiene, and health education [16–18]. This is in particular when *Ascaris* and *Trichuris* eggs remain viable in the environment for a long period of time [19]. More generally, there is a poor knowledge of the environmental contamination of STH eggs/ larvae. Although there is a consensus on how STHs are transmitted, it remains unclear where hot spots of STHs can be found in the communities, which in turn impedes both the design and the validation of water sanitation and hygiene, and health education interventions.

Indeed, studies reporting environmental contamination are mainly focused on animal STHs (e.g., Toxocara) with the ultimate goal to assess the risk for zoonotic STH transmission. These studies are conducted in countries where human STHs do not always pose a threat to public health such as Italy [20], Turkey [21], Poland [22] and Portugal [23]. Only a few studies aimed to gain insights into potential hot spots for human STHs in endemic settings, such as Cameroon [24], Kenya [25], Brazil [26], Nepal [27], Indonesia [28] and Thailand [29]. In these studies, samples were taken around households [25,29], playgrounds in school compounds [24,30], latrines [24,31] and markets [30], each of the studies highlighting the presence of STHs eggs/larvae (Ascaris: up to 27% [27], Trichuris: up to 77.0% [29], and hookworms: up to 27.5% of the samples [30]). Moreover, across these studies a plethora of methodologies have been applied for both collecting and analyzing soil samples, which makes it difficult to compare study results. For example, different amounts of soil (30-500 grams [27,28,32-37]) and soil types (clay vs. sandy and loam [38-43]) were collected at different depths (0-10 cm [32-34,37]) and at different points in time (dry vs. rainy season [28,29,33,34,42]). In general, the detection and quantification of STH eggs/larvae in soil are based on a series of steps that allow separating the STH eggs/larvae from soil particles, such as straining of soil and flotation of eggs/larvae. Yet, important differences across the currently applied methods can be noted in terms of pre-treatment (adding of surfactant [25,44] or NaOH [45]), mesh size (50–250 µm; $[\underline{21},\underline{33},\underline{35},\underline{46}]$), type and density flotation solution (FS; NaCl $[\underline{33}]$, ZnSO₄ $[\underline{30}]$, sucrose $[\underline{46}]$, $NaNO_3$ [47] and MgSO₄ [48]), centrifugal force at which flotation solution is centrifuged (1,000–2,500 rpm [30,49,50]) and time allowed for eggs to float (5–30 min [37,44,47,51,52]). Once purified, eggs/larvae are mostly detected and quantified by means of microscopic examination. Only in rare cases, DNA is extracted and subjected to STH species-specific nucleic acid amplification tests (e.g., PCR [53-55] and qPCR [56,57]). These DNA-based assays are not only considered to be more sensitive, they also allow for differentiation of the different animal and human hookworm species [55]. On top of these clear differences in methodologies, the diagnostic performance of the entire sample analysis process is mostly unknown [58], which further complicates the interpretation of the results.

In the present study, we first performed a series of laboratory experiments designed to optimize an in-house soil straining-flotation method to detect and quantify *Ascaris* and *Trichuris* eggs in soil and to validate the diagnostic performance of the optimized method when followed by both microscopy and qPCR. In a second phase, we applied this method to assess the level of STH contamination in environmental samples collected from school compounds, households and open markets in Jimma Town (Ethiopia), an area that is endemic to STHs [59,60]. Subsequently, we explored associations between the environmental contamination and both the corresponding STH epidemiology at the level of the schools and the household characteristics. Finally, we assessed the knowledge, attitude and practice (KAP) towards STH among school children.

Methods

Ethical statement

The protocols for the assessment of the environmental STH contamination, STH infections in children, household characteristics and KAP towards STH infections were approved by the Institutional Review Board of Jimma University, Ethiopia (reference numbers IHRPGD/680 and IHRPGD/466/2020). The school authorities, teachers, parents and the children were informed about the purpose and procedures of the study based on the informed consent, and adequate time was given to make an informed consent. A written informed consent was obtained from the parents/guardians. An additional separate written consent was secured from children older than 12 years. Consent forms were prepared in English, translated into the two commonly used local languages (Afaan Oromo and Amharic) and a copy was handed over to the children's parents/guardians. Only those children who were willing to participate and whose parents or guardians signed the written informed consent form were included in the study. In case of Illiteracy, a thumb print was considered as a signature.

Optimization of an in-house soil straining-flotation method

In the present study, we opted to optimize and validate our own soil straining-flotation method, based on the general laboratory procedures that are known to enhance both the purification and the concentration of STH life stages from environmental samples. Fig 1 provides a schematic overview of the different consecutive steps to detect and quantify STH eggs in soil samples applying our in-house optimized straining-flotation method. To optimize this method, we separately conducted 5 experiments during which we measured the egg recovery rate by microscopy, each building on the results from the previous experiment. We first determined the FS and the period of floating that maximized the recovery of Ascaris and Trichuris eggs (flotation experiment). Subsequently, we verified whether we could further improve the egg recovery rate by varying the centrifugal force (centrifugal force experiment) and adding detergents (Tween experiment), or whether the amount of sieved soil had an impact on the egg recovery rate (volume soil experiment). Finally, we verified whether we could further simplify the recovery of eggs from the flotation test tube without affecting the egg recovery rate (strainer experiment). For each of these spiking experiments we used cultured porcine Ascaris suum and Trichuris suis eggs. These eggs were collected from adult A. suum and T. suis, and were cultured in 2% K₂Cr₂O₇ at room temperature for at least 1.5 months to develop into infectious stages. The cultures were stored at room temperature at the Laboratory of Parasitology (Ghent University). Prior use they were neutralized by incubating them at 60°C for 1 h. The egg concentration of the two cultures were determined by examining 10 times 20 µL. The soil used in the experiments originated from Jimma Town and was sterilized at 150°C for 30 min.

Flotation experiment. A total of 24 test tubes of 15 mL was filled with different FSs, including eight tubes with 10 mL NaCl (specific gravity = 1.2), eight tubes with 10 mL MgSO₄ (specific gravity, 1.28) and eight tubes with 10 mL ZnSO₄ (specific gravity = 1.35). The number of tubes (replicates) was not based on a formal sample size calculation, rather it was based on



Fig 1. The different steps of the optimized straining-flotation method to detect and quantify STH eggs/larvae in soil. The optimized straining-flotation method consists of 9 consecutive steps, including the homogenization of the soil sample (**step 1**), sieving of the sample (**step 2**), the transfer of the sieved soil into a 50 mL tube (**step 3**), centrifugation of this test tube at 2,000 g for 5 min (**step 4**), discard supernatant, add water to bring the pellet to suspension, equally divide suspension over two 15 mL tubes, and repeat the centrifugation step (**step 5**), discard supernatants and add flotation solution to each of the test tubes (**step 6**), centrifuge the tubes at 2,000 g for 2 min (**step 7**) and put the supernatants over a PuriStrainer sieve (**step 8**), transfer the material from the PuriStrainer sieve to a microscopic slide or in a 2 mL tube for microscopic or molecular analysis, respectively (**step 9**).

https://doi.org/10.1371/journal.pntd.0010307.g001

pragmatic considerations. A total of 200 cultured eggs of porcine *A. suum* (100 eggs) and *T. suis* (100 eggs) were added to each test tube, after which the tubes were vortexed and subsequently centrifuged at 2,000 *g* for two min. Subsequently, more FS was added to form a meniscus and a cover slip was placed on top of the meniscus allowing eggs to float. To assess the impact of the period at which those eggs are allowed to float in the test tube, we removed the cover slip at different time points (5, 10, 15 and 30 min). For each time point, we performed six replicates (two replicates for each FS). In addition, we assessed the impact of serial flotation. To this end, we placed a second cover slip after removing the first and a third cover slip after removing the second. Before placing the next cover slip, we added FS to form a new meniscus. Each of these 2 additional cover slips was left for 15 min. All cover slips were placed on a microscopic slide and were examined using a compound microscope (100x magnifications). The number of *Ascaris* and *Trichuris* eggs per cover slide were recorded, resulting in a total of 72 (= 3 FSs x 8 tubes x 3 cover slips) egg counts per STH species.

Centrifugal force experiment. To assess the impact of centrifugal force on egg recovery, we determined the recovery rate of eggs when flotation test tubes were centrifuged at 1,000 and 2,000 g. To this end, 16 test tubes were filled with 10 mL of ZnSO₄ (the optimal FS derived from the flotation experiment). A total of 100 porcine *Ascaris* and 100 *Trichuris* eggs were added to ZnSO₄, after which the test tubes were centrifuged for two min at 1,000 g (eight tubes) or 2,000 g (eight tubes). Subsequently, more ZnSO₄ was added to form a meniscus and a first cover slip was placed on the meniscus allowing eggs to float. After 30 min, this cover slip was immediately placed on a microscopic slide. A second and a third cover slip were placed and removed with a 15 min interval (the optimal examination strategy derived from the flotation experiment). All cover slips were placed on a microscopic slide and were examined using a compound microscope. The number of *Ascaris* and *Trichuris* eggs per cover slide were recorded, resulting in a total of 48 (= 1 FS x 2 x centrifugal forces x 8 tubes x 3 cover slips) egg counts per STH species.

Concentration of Tween experiment. To assess the impact of detergent on egg recovery from soil we determined the recovery of eggs when the $ZnSO_4$ did include detergent. To this end, 16 test tubes were filled with 10 mL of $ZnSO_4$, with (eight tubes) or without (eight tubes) Tween 80 (0.05%). A total of 100 *Ascaris* and 100 *Trichuris* eggs were added to the $ZnSO_4$ and mixed, after which the test tubes were centrifuged for two min at 2,000 g (the optimal examination strategy derived from the centrifugal force experiment). Subsequently, more $ZnSO_4$ was added to form a meniscus and a first cover slip was placed on the meniscus allowing eggs to float. As in the centrifugal experiment, three cover slips were examined per tube, resulting in a total of 48 egg counts (2 concentrations of detergent x 1 FS x 1 x centrifugal force x 8 tubes x 3 cover slips).

Volume of soil experiment. So far, we did not add any soil to the tube. To assess the effect of volume of soil on egg recovery, we evaluated the egg recovery rate from one mL and three mL of strained soil. The volume rather than the mass of soil was preferred as it was expected that adding a volume would impact the density of the $ZnSO_4$. The choice of three mL as the highest possible volume was based on three aspects, namely 100 grams of soil corresponded with an end volume of six mL, further processing this volume would result in a reduction in the specific gravity of the $ZnSO_4$ (from 1.35 to 1.21; for three mL this is 1.28) and the need for splitting the processed soil sample into two subsamples (one for microscopic examination and one for qPCR). A total of 16 test tubes were filled with 10 mL of $ZnSO_4$ and with one mL (12 tubes) or three mL of sterilized soil (12 tubes). A total of 100 porcine *Ascaris* and 100 *Trichuris* eggs were added to the $ZnSO_4$ and mixed, after which the test tubes were centrifuged for two min at 2,000 g. Subsequently, more $ZnSO_4$ was added to form a meniscus and a first cover slip was placed on the meniscus allowing eggs to float. Similarly, like the previous experiments,

three cover slips were examined per tube $(1^{st}$ coverslip at 30 min; 2^{nd} coverslip slide at 15 min; and 3^{rd} coverslip at 15 min), resulting in a total of 72 egg counts (2 volume of soil x 1 FS x 1 x centrifugal force x 12 tubes x 3 cover slips).

Strainer experiment. From the flotation experiment it was noticed that a serial flotation is required to maximize the egg recovery rate. Because this became a time-consuming process, we verified whether we obtained similar recovery rates when the content (ZnSO₄ and eggs) of the centrifuged test tubes is strained over a small sieve, after which the filtrate is transferred to a microscopic slide. Applying this strategy would remove the repeated examination of cover slips over the course of one hour (1st coverslip removed after 30 min, 2nd coverslip at 45 min = (30 min + 15 min) and 3rd at one hour (30 min + 15 min +15 min)). To this end, 16 test tubes were filled with 10 mL of ZnSO₄ and a total of 100 *Ascaris* and 100 *Trichuris* eggs were added and mixed. All test tubes were centrifuged for two min at 2,000 g. Half of the tubes were processed as described in the previous experiments (three cover slips at 30 min, 15 min and 15 min intervals). The content of the other half of the tubes was strained through a pluriStrainer (mesh size 20 µm). The sieve was washed with tap water, after which the filtrate was aspirated using a micropipette and transferred to a microscopic slide. Finally, four cover slips were placed on slide (this was required to cover the entire aspirate), and the number of *Ascaris* and *Trichuris* were counted.

Diagnostic performance of the optimized straining-flotation method. The diagnostic performance (ability to detect at least one egg and to estimate the concentration/contamination of eggs in 100 grams of soil) of the optimized method was evaluated over a wide range of egg concentrations (10, 25, 50 and 100 eggs per 100 grams of soil). A total of 100 grams of sterilized soil were mixed with one L tap water and five mL of 0.05% Tween 80. Then, 100 porcine Ascaris eggs and 100 Trichuris eggs were added. The suspension was further homogenized for 15 min using a magnetic stirrer plate at 50 rpm. Subsequently, the suspension was strained over a tower of four sieves with different mesh sizes ($250 \,\mu$ m, $125 \,\mu$ m, $50 \,\mu$ m and $25 \,\mu$ m) ordered from top to bottom (largest mesh sizes at the top). These sieves were incorporated into the Vibratory Sieve Shaker of the Retsch GmbH machine, and were washed and shaken for 20 min at an amplitude of 0.56 mm. The filtrate on the 50 μ m and 25 μ m sieves was washed and transferred to a 50 mL test tube. These tubes were then centrifuged for 5 min at 2,000 g. Then the supernatant was removed, and the sedimented soil was equally divided into two 15 mL test tubes. Both test tubes were processed using ZnSO₄ as FS without Tween 80, applying a centrifugal force of 2,000 g for two min and using the pluriStrainer (mesh size 20 µm) to simplify the egg count process. The eggs from one tube were microscopically counted, while the eggs from the other tube were preserved at- 20°C and shipped to the Laboratory of Parasitology (Faculty of Veterinary Medicine, Ghent University, Belgium) where the samples DNAextraction was initiated. At the Microvida of the Laboratory for Medical Microbiology and Immunology, Elisabeth-TweeSteden Hospital, Tilburg (The Netherlands), DNA extraction was completed, and the detection of Ascaris and Trichuris DNA applying qPCR was performed as previously described [61-63]. For each egg concentration, this process was repeated eight times. Only half of material collected from 20 µm strainers (step 8 in Fig 1) was processed with qPCR. Processing half of the material was a strategic choice to have a back-up if the sample process failed.

Application of the optimized straining-flotation method on field samples

In total, three surveys were performed to assess the environmental contamination of STH eggs/larvae in Jimma Town (Southwest Ethiopia), including one involving 10 school compounds, one including 50 households and one conducted at nine open markets. Previous

studies highlight that STHs are highly prevalent in Jimma Town, with the prevalence in schools of any STH ranging from 8.3% to 55.0% [59,60]. In the following paragraphs we will discuss each of the surveys in more detail.

School compounds. A total of 10 governmental schools in Jimma Town were included in this survey. The selection of these schools was based on their involvement in previous STH epidemiological surveys [59,60] and drug efficacy trials [59,64]. At each school, soil samples were collected from the playgrounds, around the latrine and in the classrooms. From playgrounds, we selected the six most used playground sites and collected 6 soil samples (each 300 grams). Each sample was transferred into a polyethylene bag and labeled with both the school name and the sample collection site. Around the latrines, we collected five soil samples (each 300 grams), one from the entrance for boys, one from the entrance for girls, one in front of the latrine, one behind the latrine and one from the backyard of the latrine. Finally, six classrooms (one classroom per each of the six grades) were identified. At each classroom the entire space was swept, and 300 grams of dust/soil were transferred into a polyethylene bag. Based on this, a total of 170 soil samples (= 10 schools x (6 playground samples + 5 latrine area samples + 6 classroom samples) was collected.

Household. The environmental contamination of STH eggs/larvae was assessed at 50 households. These households represent the house of 50 school children that were randomly selected based on their infection status (27 children not excreting STH eggs and 23 children excreting STH eggs based on a single Kato-Katz thick smear) from five out of the 10 aforementioned schools. The five schools were selected based on (i) the distance between schools (only one school per kebele (the smallest unit of government administration) was included), (ii) the social-economic status of the communities they serve (schools serving both rural and urban communities were excluded), and (iii) their availability to participate in the study (schools where students had to take exams during the sample collection were excluded). The screening of the school children was part of a larger study designed to assess the impact of mass drug administration by both stool microscopy (Kato-Katz thick smear) and serology [60]. Four sites at the house of the selected school children were sampled, including the entrance of the house and latrine, the kitchen and the backyard, resulting in a total of 200 samples (50 households x 4 samples).

Open market. The mothers of the 50 randomly selected children were interviewed to determine where they buy their vegetables, fruits and cereals. Based on these interviews nine open markets were identified. At each open market, three soil samples were collected from each area where products were presented to customers (vegetables, fruits and cereals). In one big open market, 6 soil samples were collected, resulting in 30 soil samples (= 8 markets x 3 samples + 1 market x 6 samples).

Collection of soil samples. We used a shovel to collect the top layer of soil (0–3 cm). We limited the collection of soil to the superficial soil only; as we assumed that children would interact with these layers only and as previous studies have also shown that the infectious *Ascaris* and *Trichuris* eggs are not commonly found beyond this depth [65,66]. At each site, approximately 300 grams of the topsoil was collected from areas free of grass. Samples were collected during both the dry and wet season, and hence the ratio water to soil will be different over seasons (it is expected that 300 grams of soil in the wet season includes more water). To avoid contamination across sample sites, the shovel was cleaned between each sampling sites with a broom to remove the dust. In the case of classrooms, the whole classroom was first swept with a broom, after which the dust from the entire classroom was collected. All the samples were kept in a polyethylene bag that was clearly labeled with date, place and site of collection. All samples were collected in a box and transported to the Neglected Tropical Disease Laboratory of Jimma University.

Examination of the soil samples. Prior to the sample processing, we put all samples overnight at room temperature. Subsequently, samples were homogenized to ensure a more uniform distribution of eggs/larvae. Finally, a subsample of 100 grams soil was processed. We processed all the samples applying the optimized and validated straining-flotation method (see Fig 1). Briefly, 100 grams of soil were mixed with one L tap water and five mL of 0.5% Tween 80. The suspension was further homogenized for 15 min on a magnetic stirrer plate at 50 rpm. Then, the suspension was strained over a tower of four sieves with different mesh sizes $(250 \,\mu\text{m}, 125 \,\mu\text{m}, 50 \,\mu\text{m}, \text{and} \, 25 \,\mu\text{m})$ ordered from top to bottom (sieves with largest mesh size on top). These sieves were incorporated into the Vibratory Sieve Shaker of the Retsch GmbH machine and were washed and shaken at an amplitude of 0.56 mm for 20 min. The filtrate on the 50 μ m and 25 μ m was washed and transferred to a 50 mL test tube. These tubes were then centrifuged for five min at 2,000 g. Then the supernatant was removed, and the remaining soil was equally divided and transferred into two 15 mL test tubes labeled one for microscopy and the other for molecular analysis. Both test tubes were processed using $ZnSO_4$ as FS without Tween 80, applying a centrifugal force of 2,000 g for two min. Then, the supernatants of the tubes were strained separately over a pluriStrainer (mesh size 20 µm). The sieves were washed with tap water, after which the filtrate was aspirated using a micropipette. For the first tube, the filtrate was transferred to a microscopic slide and the number of Ascaris, Trichuris and hookworm-like eggs/larvae were counted. For the second tube, the supernatant was transferred into a two mL tube, which was preserved at -20°C and shipped to the Laboratory of Parasitology (Faculty of Veterinary Medicine, Ghent University, Belgium). At the Laboratory of Parasitology, the samples were prepared for DNA-extraction. At the Microvida the Laboratory for Medical Microbiology and Immunology, Elisabeth-Twee Steden Hospital, Tilburg (The Netherlands), a total of 432 DNA extraction was completed and the detection of DNA of the different STHs (differentiating Necator from Ancylostoma), Strongyloides, Taenia (differentiating T. saginata and T. solium), Enterobius vermicularis, and Hymenolepis nana applying qPCR were performed. The protocols for qPCR assays are described elsewhere [61-63].

Questionnaires. Two separate questionnaires were developed, one to obtain information on the characteristics of the households, and one on the KAP towards STH in school children. To obtain information on the household characteristics, we interviewed the mother of the household. In case the mother was not present, we interviewed the father or another head of the household. Through this questionnaire we obtained information on the age of the youngest child, presence of water at home, presence of animals in the compound, both presence and type of latrine (presence of walls, door and roof), the hygienic situation of the latrine (water in latrine, flies observed in/around latrine, visible excreta observed on latrine floor) and the defecation habits of the different family members beyond the latrine. For the KAP, we interviewed 422 school children (age 12 to 18 years old) across five out of the 10 schools. The complete questionnaires can be found in <u>S1 Info</u> (household characteristics) and <u>S2 Info</u> (KAP).

Statistical data analysis

Optimization of the straining-flotation method flotation procedures. For each experiment, we calculated the median egg recovery rate (= proportion (in %) of the 100 *Ascaris* and 100 *Trichuris* eggs recovered) across the different variables. In addition, any significant differences in egg recovery rate were assessed. Given the small sample size we used the non-parametric tests (Wilcoxon rank sum test) only. In case of multiple pair-wise comparisons, the Bonferroni correction was applied on the level of significance. In absence of a clear significant difference in egg recovery rate, either the cheapest or least labor-intensive procedure was

incorporated into the final procedure. In case there was no obvious difference in cost/labor, the procedure with the highest egg recovery rate was incorporated.

Diagnostic performance of the optimized straining-flotation method. We calculated the sensitivity (= proportion (in %) of the samples correctly classified as positive) for each of the levels of egg contamination (10 eggs / 100 grams of soil, 25 eggs / 100 grams of soil, 50 eggs / 100 grams of soil and 100 eggs / 100 grams of soil) and for the two diagnostic methods (microscopy and qPCR), separately. To explore the performance of the optimized straining-flotation method and to quantify the level of egg contamination, we determined the egg recovery rate across the different levels of egg contamination.

Application of the optimized straining-flotation method on field samples. The proportion of contaminated samples and the median level of egg/larvae contamination was determined separately for each cluster (school compounds, household, open market) and sample location (school compounds: playground, latrine, and classrooms; households: entrance of the house, entrance latrine, kitchen, and backyard; markets: vegetables, fruits and cereals). A sample was considered contaminated when either microscopy or qPCR revealed the presence of STH eggs or DNA.

To explore associations between the environmental contamination and STH infections, we first determined the correlation between the environmental contamination and the corresponding STH epidemiology at the level of the schools. To this end, we determined the Spearman correlation coefficient between the proportion of the soil samples containing STH eggs/DNA and the prevalence of STH at the school level. The data on the STH epidemiology was obtained through a study that was designed to compare both microscopy and two in-house serological assays [60], and that was completed 4–8 months prior to the assessment of the environmental contamination (October–December 2018 *vs*. March–June 2019).

Finally, we compared the test results between microscopy and qPCR. In absence of a gold standard (method that has a sensitivity and specificity of 100%), we considered the composite reference standard (CRS; [67,68]) as a proxy for a gold standard to calculate the sensitivity of the different diagnostic methods. This CRS method classified a sample as positive if eggs or DNA are found at least with one of the diagnostic methods and as negative if no eggs or DNA are found with all methods. The sensitivity for each method, as indicated by the morphology of the eggs or by the species-specific primers/probes used in the qPCR assays, except for hookworm and *Strongyloides*. For these helminths, the detection is only based qPCR (it is difficult to differentiate them from other free-living/plant parasitic worms based on their morphology only).

Questionnaires. For both questionnaires (household characteristics and KAP), we summarized the results applying standard descriptive statistics (proportions for categorical variables; median and range for continues/integer variables). We assessed whether samples of households where infected children lived were more likely to contain STH eggs/larvae. For this, we using a Chi²-test to assess difference in the presence of STH at the household level and the Wilcoxon rank sum test (two level variables) /Kruskall Wallis (variables with more than two levels) for the difference in mean number of STH eggs at the household level.

Results

Optimization of the in-house straining-flotation method procedures

<u>Table 1</u> and <u>Fig 2</u> provide a summary of the experiments designed to optimize our strainingflotation method. Using ZnSO₄ resulted in a significantly ($p \le 0.05 / 3$ pair-wise comparisons) higher egg recovery rate compared to both NaCl (W = 6.5, p = 0.009) and MgSO₄ (W = 3.5,

Experiment	Ascaris		Trichuris	
	Median egg recovery rate (%) (minimum; maximum)	<i>p</i> -value pairwise comparison	Median egg recovery rate (%) (minimum; maximum)	<i>p</i> -value pairwise comparison
Flotation expe	riment (type of flotation solution)			
$MgSO_4$ (n = 8)	7.5 (0; 21.0)	NaCl: 0.430 ZnSO ₄ : 0.003	16.5 (4; 25)	NaCl: 0.958 ZnSO ₄ : 0.226
NaCl $(n = 8)$	10.5 (3.0; 17.0)	ZnSO ₄ : 0.009	14.0 (6; 25)	ZnSO ₄ : 0.100
$ZnSO_4$ (n = 8)	23.5 (11.0; 32.0)		20.0 (12; 37)	
Flotation expe	riment (period of flotation for 1 st coverslip)			
5 min (n = 6)	0.5 (0.0; 8.0)	10 min: 0.253 15 min: 0.157 30 min: 0.044	3.0 (0; 9)	10 min: 0.868 15 min: 0.145 30 min: 0.227
10 min (n = 6)	4.0 (0.0; 10.0)	15 min: 0.625 30 min: 0.128	2.0 (1; 10)	15 min: 0.220 30 min: 0.252
15 min (n = 6)	9.0 (0.0; 12.0)	30 min: 0.470	8.0 (1; 10)	30 min: 0.809
30 min (n = 6)	9.0 (3.0; 24.0)		6.5 (1; 24)	
Flotation expe	riment (total period of flotation over three cove	erslips)		
35 min (n = 6)	3.5 (0; 30)	40 min: 0.228 45 min: 0.378 30 min: 0.378	17.5 (6; 37)	40 min: 0.688 45 min: 0.423 30 min: 0.810
40 min (n = 6)	15.0 (7; 32)	45 min: 0.872 60 min: 0.936	12.0 (7; 25)	45 min: 0.747 60 min: 0.688
45 min (n = 6)	13.5 (7; 22)	60 min: 0.810	14.5 (4; 25)	60 min: 0.423
60 min (n = 6)	13.5 (6; 24)		19.0 (6; 29)	
Centrifugal for	rce experiment			
1,000 g (n = 8)	28.0 (17.0; 55.0)	0.293	22.0 (10; 41)	0.753
2,000 g (n = 8)	48.0 (15.0; 55.0)		24.0 (14; 44)	
Tween experin	nent	1		
0% (n = 8)	48.0 (15.0; 55.0)	0.371	24.0 (14.0; 44.0)	0.010
0.05% (n = 8)	37.0 (28.0; 46.0)		39.5 (30.0; 57.0)	
Volume soil ex	periment			
1 mL (n = 12)	49.5 (13.0; 77.0)	0.665	27.5 (3; 42)	0.665
3 mL (n = 12)	51.5 (27.5; 69.0)		27.5 (6; 55)	
Strainer exper	iment	1	1	
No (n = 8)	66.5 (53.0; 80.0)	0.674	33.5 (20; 57)	0.103
Yes (n = 8)	65.5 (46.0; 78.0)		21.0 (10; 51)	

Table 1. The egg recovery rate for Ascaris and Trichuris across five experiments. The p-values were based on a non-parametric test (Wilcoxon rank sum test).

https://doi.org/10.1371/journal.pntd.0010307.t001

p = 0.003) for *Ascaris*, but not for *Trichuris*. For this STH, none of the pair-wise comparisons indicated a significant (p > 0.05 /6 pair-wise comparisons) difference across the different FS. The recovery of *Ascaris* and *Trichuris* eggs increased when the cover slip was kept for longer on the test tube. Although the pair-wise comparison did not reveal any significant differences (p > 0.05/six pair-wise comparisons) in egg recovery rate based on the first coverslip, a flotation period of 30 min did not result in any zero egg counts across both STH species (Fig 2). This was in sharp contrast with the other flotation periods where zero egg counts were observed, particularly for *Ascaris*. For this STH, zero egg counts were observed in 3, 1 and 2



Fig 2. The recovery rate of *Ascaris* and *Trichuris* eggs following serial flotation. The spaghetti plots illustrate the cumulative egg recovery rate when serial flotation was applied. The color code differentiates the time at which the first coverslip was removed from the flotation tube.

out of the six replicates for a flotation period of five, 10, and 15 min, respectively. For *Trichuris*, zero egg counts were only observed in one replicate for the flotation period of five min.

The results of the serial flotation examination of the samples indicated that repeating the flotation process increased the final egg recovery rate for both *Ascaris* and *Trichuris*, and this regardless of the period of flotation of the first cover slip. There was no significant (p > 0.05/6 pair-wise comparisons) difference in the total number of eggs recovered between a serial flotation of 35 min (= 5 + 15 + 15), 40 min (= 10 + 15 + 15), 45 min (= 15 + 15 + 15) and 60 min (= 30 + 15 + 15).

The median egg recovery rates in the remaining laboratory experiments did not reveal major differences in egg recovery rate. A difference in median egg recovery rates of more than 10-point percent was observed between the centrifugal forces for *Ascaris* (2,000 g: 48.0% vs. 1,000 g: 28.0%), the Tween 80 concentrations for both *Ascaris* (Tween: 37.0% vs. no Tween: 48.0%) and *Trichuris* (Tween: 39.5% vs. no Tween: 24.0%), and when the serial flotation step was replaced by an additional straining step for *Trichuris* (no straining: 33.5% vs. straining: 21.0%). The non-parametric test confirmed that adding Tween resulted in a significantly improved egg recovery rate (W = 8, p = 0.010) for *Trichuris*. In all other comparisons, the test did not reveal any significant difference in egg recovery rate (p > 0.05).

Based on the outcome of the experiments we decided to use $ZnSO_4$ as FS, to centrifuge the tubes at 2,000 g and to replace the repeated flotation step by an additional straining step.

Concerning the Tween, we decided not to add the Tween into FS, rather we added 0.5% Tween 80 to the soil to facilitate homogenization of the eggs/larvae prior to sieving. Fig 1 provides a schematic overview of our optimized in-house straining-flotation method.

Diagnostic performance of the optimized straining-flotation method

Combining both microscopy and qPCR resulted in an overall sensitivity of 81.3% for Ascaris and 87.5% for Trichuris. The outcome of the different experiments is graphically summarized in Fig 3. S1 Table provides a numeric summary of the diagnostic performance. For microscopy, the sensitivity was 100% when the concentration was 50 eggs per 100 grams of soil for both STHs. For a concentration of 25 eggs per 100 grams of soil, the sensitivity was 75.0% (95%CI: 45.0; 100) for Ascaris and 62.5% (95CI: 29.9; 96.0) for Trichuris, while for a concentration of 10 eggs/100 grams of soil the sensitivity was 25.0% (95%CI: 0.0; 55.5) and 12.5% (95% CI: 0.0; 35.4), respectively. Given the observed recovery rate in the experiments designed to optimize the straining-flotation method, it did not come with a surprise that the estimated egg concentration is underestimating the true underlying egg contamination. Yet, it is important to note that the egg recovery rate did not remain unchanged across the different levels of egg contamination for Ascaris but increased as a function of increasing egg contamination. For Trichuris, the egg recovery rates remain largely unchanged. Generally, qPCR was less sensitive than microscopy for Ascaris (75.0% vs. 59.4%). For Trichuris, qPCR was more sensitive in a higher detection rate when concentration was low (10 eggs: 50.0% vs. 12.5%; 25 eggs: 87.5% vs. 62.5%) but resulted in a lower rate when concentration was high (50 eggs: 87.5% vs. 100%; 100 eggs: 87.5% vs. 100%).

Application of the optimized straining-flotation method on field samples

The analysis of the 399 field samples (schools: 170; households: 199; markets: 30) with both microscopy and qPCR revealed the presence of 8 different helminth species of medical importance, including but not limited to the human STHs. The most prevalent helminths were *Ascaris* (65.7%) and *Taenia saginata* (52.4%), followed by *Trichuris* (21.6%). The other helminths were only observed in a minority of the cases (*Enterobius*: 5.8%; *Hymenolepis*: 5.3%; *Strongyloides*: 1.5%; *Necator*: 0.8%; *Schistosoma*: 0.8%). A numerical overview of environmental contamination for all helminth species at the different school compounds, households and open markets is provided in S2–S4 Tables. In the following paragraphs we will focus on the environmental contamination of STHs only.

The environmental STH contamination at school compounds. The environmental contamination with human STHs at school compounds is illustrated in Fig 4. Overall, human STHs were observed in 113 out of the 170 (66.5%) samples that were examined by both microscopy and qPCR. *Ascaris* was the most observed STH species (61.2%), followed by *Trichuris* (17.1%). Of the two human hookworms *N. americanus* was observed in 2 samples (1.2%) only. Both *Ascaris* and *Trichuris* were found across all ten schools, the sample prevalence ranging from 35.3% to 94.1% for *Ascaris*, and from 5.9% to 41.2% in school #6 for *Trichuris*. *N. americanus* was only observed in school #1 (11.8%) (Fig 4A).

When focusing on the different sampling sites within the school compounds (Fig 4B), the sample prevalence of any human STH was the highest for classrooms (81.6%), followed by both latrines (62.0%) and playground (55.0%). The sample prevalence for each of the three STHs followed a similar pattern, sample prevalence being highest for classrooms, followed by latrines and playground (*Ascaris*: 73.3% *vs*. 60.0% *vs*. 50.0%; *Trichuris*: 26.7% *vs*. 12.0% *vs*. 11.7%; *N. americanus*: 3.3% *vs*. 0.0% *vs*. 0.0%). Around the latrines (Fig 4C), the sample prevalence of any human STH was at least 50% and did not vary across the different sampling sites



Fig 3. The diagnostic performance of the optimized straining-flotation method to detect and quantify *Ascaris* and *Trichuris* eggs in soil. The panels describe the sensitivity for both microscopy and qPCR (top row panels), the egg recovery rate for microscopy (middle row panels) and the Ct-values for qPCR (bottom row panels) separately for *Ascaris* (left column panels) and *Trichuris* (right column panels).

(behind the latrine: 50.0%; the back yard of the latrine: 50.0%, in front of the latrine: 70.0%; the entrance for girls: 70.0%; the entrance for boys: 70.0%). While for *Ascaris*, the sample prevalence did not considerably vary across the sampling sites (50.0% to 70.0%), the presence of *Trichuris* was not observed in samples from both behind and in front of the latrine. The sample



Fig 4. The environmental STH contamination at 10 school compounds, Jimma Town (Ethiopia). This figure illustrates the environmental soiltransmitted helminth contamination across 10 governmental schools in Jimma Town, Ethiopia (**Panel A**), at different sites within schools (classroom, latrine and playground; **Panel B**), around different places around the school latrine (**Panel C**), and the association between prevalence of STH infections in children and the proportion of the samples contaminated with STH at the school level (**Panel D**).

prevalence at both the entrance for girls and the backyard of the latrine equaled 10.0%, and at the male entrance *Trichuris* was found in 40.0% of the samples. The association between prevalence of STH infections in children and the proportion of the samples contaminated with STH at the school level is illustrated in Fig 4D, highlighting that the prevalence of any infections in school children was significantly associated with the proportion of environmental samples containing any STH life stages (Rs = 0.42, p = 0.02). S5 Table provides an overview of the STH infections across school children in the 10 schools in Jimma Town, Ethiopia.

The environmental STH contamination at households. The environmental contamination with human STHs at the households is illustrated in Fig 5. Overall, human STHs were observed in 147 out of the 199 (73.9%) samples that were examined by both microscopy and qPCR. Similarly, as for the school compounds, *Ascaris* was the most observed STH species (71.4%), followed by *Trichuris* (23.6%). *N. americanus* was again observed in 1 sample (0.5%). With exception of one, *Ascaris* was found in all households, *Trichuris* was found in 60.0% of the households (Fig 5A). *Necator* was only observed in one household. The contamination varied across the different sampling sites (Fig 5B). Generally, samples collected at the latrine entrance were most contaminated (any STH: 84.0%; *Ascaris*: 82.9%; *Trichuris*: 30.0%). The





least contamination was observed in the kitchen (any STH: 65.3%; *Ascaris*: 59.2%; *Trichuris*: 18.4%).

Table 2 summarizes the characteristics of the different households and the corresponding environmental contamination measured by the mean STH egg counts across the samples. On average, 5.5 people lived in the households. In 12 out of the 50 households, at least one child <3 years of age was reported and in 40 households (80.0%) animals (e.g. dogs, cats, chicken and cattle) were observed. With only a few exceptions (n = 2; 4.0%), all households had a latrine. Most of the latrines were open well (n = 34; 68.0%), all others were pit latrines (n = 14;28.0%). Half of the latrines did not have a wall (n = 26; 52.0%) or a roof (n = 25, 50.0%). In one third of the cases, there was no door (n = 15; 30.0%). Flies were observed around the latrines in 34 households (68.0%), and excreta were visible in six households (12.0%). The 20% of the household heads reported that their children practiced open field defecation. Almost all visited households (98.0%) did not have water in the latrine and 46.0% did not have any water supply at their household compounds. Most of the kitchens (n = 36, 72.0%) had walls and a floor of soil, and 11 households had an open space kitchen (n = 11, 22.0%). In only in a few cases (n = 3; 6.0%), we observed a kitchen with walls and a concrete floor. The mean distance from the latrine entrance to the house entrance was 16.7 m (ranging from 4 to 35 m). Given the high prevalence of STHs at the household (presence of STH eggs/DNA was observed in at least 2

Table 2. The characteristics and the corresponding environmental contamination of 50 households in Jimma Town, Ethiopia. The environmental contamination measured by the mean soil-transmitted helminth egg counts across the samples. The *p*-value was based on the Wilcoxon rank sum test in case of two-level variables, while in case of variables with more than two levels it was based on the Kruskall Wallis test.

Variable	N (%)	Median STH egg count (minimum; maximum	<i>p</i> -value			
Presence of child <3years old						
Yes	12 (24.0)	12.8 (1.5; 441.5)	0.400			
No	38 (76.0)	9.0 (0.0; 1,153.0)				
Presence of animal (dog, cat, hen and	d cow)					
Yes	40 (80.0)	12.8 (5.5; 34.0)	0.149			
No	10 (20.0)	7.5 (0.0; 1,153.0)				
Presence of a latrine						
Yes	48 (96.0)	9.8 (0.0; 1,153.0	0.921			
No	2 (4.0)	221.0 (0.5; 441.5)				
Type of latrine						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.826			
Pit latrine	14 (28.0)	7.8 (0.0; 34.0				
Open well	34 (68.0)	10.0 (0.0; 1,153.0)				
Wall around latrine						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.943			
Yes	26 (52.0)	10.3 (0; 49.0)				
No	24 (48.0)	9.5 (1.0; 1,153.0)				
Latrine door						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.986			
Yes	15 (30.0)	10.5 (0; 34.0)				
No	33 (66.0)	9.5 (0; 1,153.0)				
Latrine roof						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.799			
Yes	25 (50.0)	10.0 (0.0; 42.5)				
No	23 (46.0)	9.5 (1.0; 1,153.0)				
Flies around latrine						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.923			
Yes	32 (64.0)	9.8 (0.0; 1,153.0)				
No	16 (32.0)	8.3 (0.0; 34.0)				
Visible stool observed on household	areas					
Yes	6 (12.0)	9.5 (0; 1,153.0)	0.881			
No	44 (88.0)	10.3 (2; 14.5)				
Defecation outside latrine		1				
Yes	12 (24.0)	8.0 (0.5; 441.5)	0.503			
No	38 (76.0)	9.8 (0.0; 1,153.0)				
Water supply at home		1	1			
Yes	27 (54.0)	10.0 (0.0; 1,153.0)	0.311			
No	23 (46.0)	9.5 (0; 441.5)				
Water in the latrine						
No latrine	2 (4.0)	221.0 (0.5; 441.5)	0.342			
Yes	1 (2.0)	34.0 (34.0; 34.0)				
No	47 (94.0)	8.25 (0.5; 16.0)				
ſype of kitchen						
Open space	11 (22.0)	9.5 (0.0; 30.5)	0.607			
Soil floor with wall and roof	36 (72.0)	10.3 (0; 1,153)				
Cement floor with wall and roof	3 (6.0)	5.0 (3.5; 34.0)				

https://doi.org/10.1371/journal.pntd.0010307.t002

samples in 43 out of 49 households), we did not explore any associations between presence of environmental contamination at the household level and the household characteristics and the infection status of the child. There were no household characteristics that explained the differences in eggs in 100 grams of soil (*p*-value >0.05). The median (minimum; maximum) environmental contamination in the households of the children with STH infections equaled 10.5 eggs per 100 grams of soil (0.5; 1,153.0) and was not significantly (p = 0.453) different from the households of children without STH infections (6.5 (0.0; 441.5)).

The environmental contamination with human STHs at the open markets. The environmental contamination with human STHs at the open markets is illustrated in Fig 5C and 5D. Overall, human STHs were observed in 20 out of the 30 samples that were examined by both microscopy and qPCR. Here too, *Ascaris* was the most observed STH species (53.3%), followed by *Trichuris* (33.3%). *Necator* was not observed in any of the markets. *Ascaris* was detected at seven out of the nine open markets and *Trichuris* at six open markets. At most open markets, at least two out of the three (eight markets) or six soil samples (one large open market) were contaminated with any STHs (seven out of nine markets) and *Ascaris* (six out of nine markets). For *Trichuris*, majority of the markets (n = 5) had less than two contaminated samples. There was little variation in contamination with any STHs across the locations where fruit (60.0%), grains (70.0%) and vegetables (70.0%) where sold. Yet, on areas where vegetables were sold, mainly *Ascaris* (70.0%) was observed and to a lesser extent (*Trichuris*). At the other areas, both STHs were found at a similar rate.

Comparison of microscopy and qPCR. When the optimized method was applied to field samples, eggs and/or DNA of *Ascaris* and *Trichuris* was detected in 262 and 86 field samples, respectively. Assuming a perfect specificity for both microscopy and qPCR, the sensitivity of microscopy was 77.9% for *Ascaris* and 38.4% for *Trichuris*. Compared to microscopy, qPCR was less sensitive for *Ascaris* (61.0%) and more sensitive for *Trichuris* (74.4%). As illustrated in S1 Fig, the Ct-value dropped as a function of increasing numbers of eggs counted by microscopy for *Ascaris* and to a lesser extent for *Trichuris*. For the other helminths of medical importance, qPCR outcompeted microscopy. For example, based on the results of microscopy and qPCR and 6.2% by microscopy. Similar findings were observed for *Enterobius* (23 cases, of which 91.3% by qPCR and 59.1% by microscopy). For the remaining helminth species, a comparison was difficult as morphological differentiation was not straight forward (*Strongyloides* and hookworms) or due to the low number of cases (*Schistosoma*).

KAP towards STHs

A total of 422 school children participated in this questionnaire. The age of the children ranged from 12 to 18 years (median = 14), and the ratio boys to girls equaled 1:1.21. Half of the children were Muslim (51.9%), the other half was Christian (48.1%). The education of the parents was poor in majority of the cases. More than half of the fathers did not complete senior high school (illiterate: 13.7%; elementary school: 25.8%; junior high school: 18.0%) and half of the mother did not complete junior high school (illiterate: 28.0%; elementary school: 30.6%). Majority of the fathers were either farmer (n = 125; 29.6%) or employees of the government (n = 122; 28.9%), while mothers were mainly a housewife or a cleaner (n = 287; 68.0%).

Generally, most of children were aware of STHs, 372 (88.2%) children confirming that they had heard the STH-related words, such as 'intestinal worms', '*Ascaris*', '*Trichuris*' and '*hook-worms*', before. This was much in line with other major diseases (HIV: 98.3%; malaria: 91.0% and tuberculosis: 82.0%). However, important differences across the words were observed. For

example, 'intestinal worms' and 'Ascaris' were known by approximately 90% of the children. The words 'hookworms' and 'Trichuris' were only known by 12.6% and 0.9% of the children, respectively. Teachers were the main source of information (88.9%), followed by television (73.7%) and radio (63.3%). Children also heard about the aforementioned words through nurses and doctors (26.8%) and health extension workers (25.8%). On the impact of STHs on the human body, majority confirmed that STHs cause stomach-ache (n = 384; 91.0%) and that they do share food of the host (n = 315, 74.6%). Approximately half of the students (47.2%) did not know that STHs (hookworm) also suck blood. A minority of the children falsely declared that STHs cause blood in the urine (19.7%) and blindness (10.4%). Many students (82.9%) knew that adult STHs are dwelling in the intestines, but there was also a minority of the children that thought that adult STHs can be found in other parts of the human body (skin: 15.2%, lung: 13.0%, heart: 7.1%, head: 5.9% and eyes: 4.0%). Only few students (n = 26, 6.2%), did not know that doctors and nurses can provide drugs against STH infections. In addition, a considerable part of the students (n = 114, 27.0%) incorrectly answered that there is a vaccine against STH infections. Although, most of the students know that STHs are not transmitted through air (n = 95, 22.5%), insect bites (n = 58, 13.7%), blood (n = 64, 15.2%) and saliva (n = 27, 6.4%), only 173 children (41.0%) confirmed that STHs are transmitted through stool. Many children, 327 (77.5%) reported that STH is transmitted by eating raw meat. Children usually defecate at the latrine of their home 64.0% (n = 270), followed by the latrine in the compound (village common latrine; n = 145; 34.4%). Only few students defecate in the latrines of the school (0.7%) and in the bush (0.9%). Majority of the children reported that they always washed their hands before eating food (n = 400; 94.8%) and after using the latrine/toilet (n = 361; 85.5%). Washing soap was often used (n = 255; 60.4%). From all the interviewed students, 18.5% (n = 78) did not trim their fingernails. Most of the children (n = 292; 69.2%) trimmed their fingernails once a week, 11.1% (n = 47) once every two weeks and 1.7% (n = 7) less than one time per month. Almost all students (n = 419; 99.3%) wore shoes, but 21.8% (n = 92) of them did not always do this. From all the study participants, 33.1% (n = 135) did not worry about the risk of getting an STH infection and 11.9% (n = 49) would not be anxious if they got infection with intestinal worms.

Discussion

Large-scale deworming programs have been successful in reducing the disease burden attributable to STH infections, but re-infection in the absence of other intervention measures is unavoidable. Today, it remains unclear where and how (re-)infection occurs, which further impedes the development of targeted and perhaps more cost-effective control strategies. In the present study, we aimed to identify possible sources of infection by assessing the environmental contamination in school compounds, households and open markets. To this end, we first performed a series of laboratory experiments designed to optimize an in-house straining-flotation method to detect and quantify *Ascaris* and *Trichuris* eggs in soil and to validate the diagnostic performance of the optimized method when followed by both microscopy examination and qPCR. In a second phase, we applied this method to assess the level of STH contamination in environmental samples collected from school compounds, households and open markets in an area that is endemic to STHs. Subsequently, we explored associations between the environmental contamination and both the corresponding STH epidemiology at the level of the schools and the household characteristics. Finally, we assessed the KAP towards STH among school children.

Our soil straining-flotation method has a moderate diagnostic performance

Our experiments were designed to optimize our in-house soil straining-flotation method indicated that (i) $ZnSO_4$ (specific density = 1.35), (ii) 30 min of flotation, (ii) increasing centrifugal force and (iii) examination of multiple coverslips following repeated flotation significantly improved the recovery of *Ascaris* and *Trichuris* eggs. These findings were not unexpected and are in line with previous studies that assessed the impact of both the specific density and the nature of FS [69,70], and repeated flotation [25]. In our final standard operating procedure (SOP; Fig 1), we replaced the repeated flotation step by an additional straining step to reduce the sample process time to 30 min per sample when samples were examined by microscopy. Generally, the median reported time in literature equaled 2.0 h (ranging from <1 h to 2 days) [58], highlighting that our method allows for a moderately high sample throughput.

For microscopy, the analytical sensitivity (= concentration of eggs for which a positive test result was obtained in at least 95% of the replicates) of our final SOP was 50 eggs per 100 grams of soil (or 1 egg in 2 grams of soil) for both STHs (*Ascaris* and *Trichuris*). The median egg recovery rate for 100 eggs per grams of soil was 36.0% for *Ascaris* and 8.0% for *Trichuris*. For qPCR, the analytical sensitivity was 100 eggs per 100 grams of soil for *Ascaris*, while for *Trichuris* this was >100 eggs per 100 grams of soil. This difference in analytic sensitivity might be due to a difference in the amount of soil that was analyzed by microscopy and qPCR. While we analyzed all soil that was collected from the 20 µm strainer with microscopy (**step 9** in Fig 1), only half of this was analyzed with qPCR. Processing half of the material was a strategic choice to have a back-up if the sample process failed.

Although a head-to-head comparison with other studies is difficult to impossible due the diversity in SOPs and different helminth species (see Introduction), two important aspects can be noted. First, we stress-tested our method in a range of egg counts (0.1 to 1 egg per gram of soil) and mass of soil (100 grams) that have been rarely examined. Indeed, based on a review of the literature the median egg concentration was 20.0 eggs per gram of soil (ranging from 4 to 32,393) for Ascaris [58]. For Trichuris, the egg concentration ranged from 1 to 32,393 eggs per gram of soil for *Trichuris* (only two studies were reported in this review; [58]). Compared to the few studies available, our method revealed to have a low analytic sensitivity. For example, spiking experiments with Toxocara canis eggs (1, 10, 25, 50, 100, and 200) into one gram of soil [69], revealed that at least one egg was microscopically detected in replicates with 10 eggs, resulting in an analytical sensitivity not lower than 10 eggs per gram soil (vs. 0.5 eggs per gram of soil in the present study). Second, the recovery rate of our method revealed to be within the range reported for Ascaris (median egg recovery rate reported by [58] equals 25.0%, ranging from 9.5 to 82%), but at the lower end for Trichuris (only two studies were reported in this review; 80.0% and 96.4%; [58]). Although, one cannot rule out variation in soil type or SOPs, this poor egg recovery rate for Trichuris can also be explained by the mesh size we used to strain the soil samples (25 µm). Trichuris eggs measure size 50–55 μ m x 20–25 μ m [71], and therefore they may pass the sieve resulting in the observed poor egg recovery rate. Including a sieve with a mesh size of 20 µm would be an obvious modification of our SOP.

Although the aforementioned paragraphs highlight that our method allows for a moderately high throughput and has a moderate diagnostic performance, the purchase of the 'Vibratory Sieve Shaker' (with the corresponding sieves) is a substantial investment (8,000 EURO, incl 17% VAT). In addition to this, the access of constant flow of relatively large volumes of water, was also perceived as an important obstacle.

Life stages of helminths of medical important helminths are ubiquitous in the environment

Our result highlights that the presence of 8 different helminth species of medical importance, including but not limited to the human STHs. The most ubiquitous helminths were *Ascaris* (65.7%) and *Taenia saginata* (52.4%), followed by *Trichuris* (21.6%). The other helminths were only observed in a minority of the cases (*Enterobius*: 5.8%; *Hymenolepis*: 5.3%; *Strongyloides*: 1.5%; *Necator*: 0.8%; *Schistosoma*: 0.8%). Generally, both the presence and the prevalence of these helminths in the environment are reflecting the infection rates of school children [59,60] and adults [60], except for *Taenia*. For this helminth, the infection rates 2.3% [72–73] in Ethiopia (including our study area) are lower than when one would expect based on the environmental contamination. This might be explained by both the poor sensitivity of the method used to diagnose infections (Kato-Katz thick smear: 52% (95% credible interval [7–94]) *vs*. qPCR sensitivity 94% (95% credible interval [88–98]) [74] and the number of eggs excreted in the environment once a gravid proglottid has actively left the host (one gravid proglottid may contain up to 2,000 eggs [75]).

The environmental STH contamination was high across all major sample locations, households being the most contaminated (73.9% *vs.* ~66% in both open markets and school compounds). The contamination level at school compounds was exceeding that previously reported (Nigeria: 53.6% [76] and 32.0% [77]; Cameroon: 17% [78]; Ethiopia: 11.3% [79]). The contamination at households was comparable to that observed in a study in Bangladesh (78.0% [24]) but higher that of a study in Kenya 26.8% [25]. The contamination level in the present study was higher than that previously reported (15% in Cameron [80]).

Although a head-to comparison remains once more difficult, this higher prevalence STH in our surveys could be due to the higher amount of soil (100 grams of soil) and combining both microscopy and qPCR in the sample process. The latter method allowing for a more sensitive (though not for all helminth species), specific (e.g., speciation of the hookworm and *Taenia* species) and simultaneous detection of multiple helminth species [55,56]. Although these results indicate that helminths are ubiquitous in the environment, we did not assess the viability of the different life stages. Viability can be assessed by specific dyes [81,82]. Because of this, we are not able to accurately estimate the risk of infection.

Are schools a source of STH infections in children or are children contaminating schools?

There was an association between prevalence of STH infections in children and the proportion of samples contaminated with STHs at the school level (see Fig 5D). Yet, it remains unclear whether schools are a source of STH infections or whether children are contributing to school contamination. In either way, the fact that the contamination in classrooms is higher than around latrines is remarkable, yet not unexpected. This is because, the inside of the latrines was extremely contaminated with human stool. As a consequence of this, it is not unlikely that children step into human stool when using the latrines and stepping into the classroom with shoes contaminated with stool. Moreover, while life stages surrounding the latrines might be washed away during episodes of rain, this is not likely to occur in the roofed classrooms. Lack of awareness on route of STH transmission and prevention methods among study participants further highlight the need for health education and improved WASH intervention in the schools to minimize both environmental contamination and transmission of STH.

Other limitations of the study

This study has a number of other limitations. First, while the environmental samples were collected during both the dry and wet season, we did not dry the samples prior to processing them with our in-house straining flotation method. This might be important to for future studies designed to assess seasonal differences in environmental contamination. Second, to gain more insights in the role of children in contaminating the school, it would have been important to verify an association both between children defecating at the school and the environmental contamination found at schools and between defecating at school and the STH infection status. Although we indeed capture the defecating behavior in the KAP survey, due to time constrains (survey was conducted shortly before school holidays) it was assessed in five out of ten schools only. Moreover, the children were unexpectedly dewormed as part of the national STH control program while the KAP survey was conducted, and hence we were not able to verify associations between their KAP towards STH and their infection status. Finally, the validation of our in-house straining-flotation method was limited to *Ascaris* and *Trichuris* only. It would be important to also assess this for the other helminth species.

Conclusions

Our optimized straining-flotation method has a moderate diagnostic performance and revealed that life stages of helminths are ubiquitous in the environment, reflecting both the poor sanitary facilities at the households and schools, and the lack of awareness of STHs. Further research is required to gain more insights into the contribution of these life stages to transmission.

Supporting information

S1 Fig. The association in contamination in environmental samples measured by microscopy and qPCR. These scatterplots represent the association in *Ascaris* and *Trichuris* contamination measured by microscopy (in eggs per 100 grams of soil) and qPCR (in Ct). A Ct-value of zero indicates absence of DNA. (TIF)

S1 Table. The ability of the optimized straining-flotation method to detect and quantify STH eggs in soil. The egg counts were adjusted for the amount of sample examined. The Ct-value summarized in this table only reflect those that were observed in qPCR-positive samples only.

(DOC)

S2 Table. Helminth contamination of soil samples collected in 10 school compounds in Jimma Town, Ethiopia.

(DOC)

S3 Table. Helminth contamination of soil samples collected from 50 households at Jimma Town, Ethiopia.

(DOC)

S4 Table. Helminth contamination of soil samples collected from 9 markets at Jimma Town, Ethiopia. (DOC)

S5 Table. Soil-transmitted helminths. infections in 10 schools in Jimma Town, Ethiopia. (DOC)

S1 Info. The questionnaire to gain insights into the characteristics of the households. (DOC)

S2 Info. The questionnaire on knowledge, attitude and practice towards soil-transmitted helminths.

(DOC)

S1 Data. The optimization and the diagnostic performance of the soil straining-flotation method.

(XLSX)

S2 Data. The environmental contamination at school compounds, households and open markets.

(XLSX)

S3 Data. The household characteristics. (XLSX)

S4 Data. The knowledge, attitude and practice towards soil-transmitted helminths. (XLSX)

Acknowledgments

We would like to thank the children and their parents and the teachers for their participation in this study.

Author Contributions

Conceptualization: Bamlaku Tadege, Zeleke Mekonnen, Eskindir Loha, Stijn Casaert, Bruno Levecke.

Data curation: Bamlaku Tadege, Mio Ayana, Bruno Levecke.

Formal analysis: Bamlaku Tadege, Mio Ayana, Bruno Levecke.

Funding acquisition: Zeleke Mekonnen, Bruno Levecke.

Investigation: Bamlaku Tadege, Daniel Dana, Bizuwarek Sharew, Eden Dereje, Jaco J. Verweij, Stijn Casaert, Mio Ayana, Bruno Levecke.

Project administration: Zeleke Mekonnen, Bruno Levecke.

Supervision: Zeleke Mekonnen, Eskindir Loha, Johnny Vlaminck, Bruno Levecke.

Visualization: Johnny Vlaminck.

Writing - original draft: Bamlaku Tadege.

Writing – review & editing: Zeleke Mekonnen, Daniel Dana, Bizuwarek Sharew, Eden Dereje, Eskindir Loha, Jaco J. Verweij, Stijn Casaert, Johnny Vlaminck, Mio Ayana, Bruno Levecke.

References

- Hotez PJ, Fenwick A, Savioli L, Molyneux DH. Rescuing the bottom billion through control of neglected tropical diseases. Lancet. 2009; 373(9674): 1570–1575. https://doi.org/10.1016/S0140-6736(09) 60233-6 PMID: 19410718
- Bethony J, Brooker S, Albonico M, Geiger SM, Loukas A, Diemert D, et al. Soil-transmitted helminth infections: ascariasis, trichuriasis, and hookworm. *Lancet*. 2006; 367(9521):1521–32. <u>https://doi.org/ 10.1016/S0140-6736(06)68653-4</u> PMID: 16679166

- Traversa D, Frangipane di Regalbono A, Di Cesare A, La Torre F, Drake J, Pietrobelli M. Environmental contamination by canine geohelminths. *Parasit Vectors*. 2014; 7(67). <u>https://doi.org/10.1186/1756-</u> 3305-7-67 PMID: 24524656
- Jourdan PM, Lamberton PHL, Fenwick A, Addiss DG. Soil-transmitted helminth infections. *Lancet.* 2018; 391(10117):252–265. https://doi.org/10.1016/S0140-6736(17)31930-X PMID: 28882382
- GBD 2017 DALYs and HALE Collaborators. Global, regional, and national disability-adjusted life-years (DALYs) for 359 diseases and injuries and healthy life expectancy (HALE) for 195 countries and territories, 1990–2017: a systematic analysis for the global burden of disease study 2017. *Lancet*. 2018; 392 (10159):1859–1922. https://doi.org/10.1016/S0140-6736(18)32335-3 PMID: 30415748
- Brooker S, Clements AC, Bundy DA. Global epidemiology, ecology and control of soil-transmitted helminth infections. *Adv Parasitol.* 2006; 62:221–61. <u>https://doi.org/10.1016/S0065-308X(05)62007-6</u> PMID: 16647972
- World Health Organization. Eliminating soil-transmitted helminthiases as a public health problem in children: progress report 2001–2010 and strategic plan 2011–2020 WHO Press, Geneva (2012) (World Health Organisation)
- 8. World Health Organization. Soil-transmitted helminth infections WHO Fact Sheet N°366 (2015)
- Drake L, Jukes M, Sternberg RJ and Bundy DAP. Geo-helminthiasis (ascariasis, trichuriasis and hookworm): Cognitive and development impact. Seminars in Semin Pediatr Infect Dis. 2000; 11(4): 245–251.
- Shang YU, Tang LH, Zhou SS, Chen YD, Yang YC, Lin SX. Stunting and soil-transmitted-helminth infections among school-age pupils in rural areas of southern China. *Parasit Vectors*. 2010; 3(1):1–6. https://doi.org/10.1186/1756-3305-3-1 PMID: 20051120
- 11. Miguel E, Kremer M. Worms: identifying impacts on education and health in the presence of treatment externalities. *Econometrica*. 2003; 72 (1): 159–217
- 12. World Health Organization. Preventive chemotherapy to control soil-transmitted helminth infections in at-risk population groups; WHO, Geneva, 2017
- World Health Organization. 2030 targets for soil-transmitted helminthiases control programmes. WHO 2019.
- Jourdan PM, Montresor A, Walson JL. Building on the success of soil-transmitted helminth control the future of deworming. *PLoS Negl Trop Dis.* 2017; 11(4): e0005497. https://doi.org/10.1371/journal.pntd. 0005497 PMID: 28426784
- Marocco C, Bangert M, Joseph SA, Fitzpatrick C, Montresor A. Preventive chemotherapy in one year reduces by over 80% the number of individuals with soil-transmitted helminthiases causing morbidity: results from meta-analysis. *Trans R Soc Trop Med Hyg.* 2017; 111(1):12–7. <u>https://doi.org/10.1093/</u> trstmh/trx011 PMID: 28340144
- Ziegelbauer K, Speich B, Mäusezahl D. Effect of sanitation on soil-transmitted helminth infection: systematic review and meta-analysis. *PLoS Med.* 2012; 9: e1001162. <u>https://doi.org/10.1371/journal.pmed.1001162</u> PMID: 22291577
- Anderson R, Truscott J, Hollingsworth TD. The coverage and frequency of mass drug administration required to eliminate persistent transmission of soil-transmitted helminths. *Philos Trans R Soc Lond B Biol Sci.* 2014, 369. https://doi.org/10.1098/rstb.2013.0435 PMID: 24821921
- Truscott J, Hollingsworth T, Brooker S, Anderson R. Can chemotherapy alone eliminate the transmission of soil transmitted helminths? *Parasit Vectors*. 2014; 7:266. <u>https://doi.org/10.1186/1756-3305-7-266 PMID: 24916278</u>
- Nordin A, Nyberg K, Vinnerâs B. Inactivation of Ascaris eggs in source-separated urine and feces by ammonia at ambient temperatures. Appl Environ Microbiol. 2009; 75:662–7. <u>https://doi.org/10.1128/</u> AEM.01250-08 PMID: 19060175
- Habluetzel A, Traldi G, Ruggieri S, Attili AR, Scuppa P, Marchetti R, et al. An estimation of *Toxocara canis* prevalence in dogs, environmental egg contamination and risk of human infection in the Marche region of Italy. *Vet Parasitol.* 2003; 113(3–4):243–252. https://doi.org/10.1016/s0304-4017(03)00082-7 PMID: 12719139
- Avcioglu H, Burgu A. Prevalence of Toxocara ova in soil samples from public parks in Ankara Turkey. Vector Borne Zoonotic Dis. 2008; 8 (3): 345–350. https://doi.org/10.1089/vbz.2007.0212 PMID: 18494602
- 22. Bojar H, Kłapeć T. Contamination of soil with eggs of geohelminths in recreational areas in the Lublin region of Poland. *Ann Agric Environ Med.* 2012; 19(2):267–70. PMID: 22742799
- Otero D, Alho AM, Nijsse R, Roelfsema J, Overgaauw P, Madeira de Carvalho L. Environmental contamination with *Toxocara* spp. eggs in public parks and playground sandpits of Greater Lisbon, Portugal. *J Infect Public Health.* 2018; 11(1):94–98. https://doi.org/10.1016/j.jiph.2017.05.002 PMID: 28545900

- Nkouayep VR, Ngatou Tchakounté B, Wabo Poné J. Profile of geohelminth eggs, cysts, and oocysts of protozoans contaminating the soils of ten primary schools in Dschang, West Cameroon. *J Parasitol Res.* 2017; 1534675, https://doi.org/10.1155/2017/1534675 PMID: 29238615
- Steinbaum L, Njenga SM, Kihara J, Boehm AB, Davis J, Null C, et al. Soil-transmitted helminth eggs are present in soil at multiple locations within households in rural Kenya. *PLoS One*. 2016; 11(6): e0157780. https://doi.org/10.1371/journal.pone.0157780 PMID: 27341102
- 26. Sprenger LK, Green KT, Molento MB. Geohelminth contamination of public areas and epidemiological risk factors in Curitiba, Brazil. *Rev Bras Parasitol Vet.* 2014; 23(1):69–73. https://doi.org/10.1590/ s1984-29612014009 PMID: 24728363
- Rai SK, Uga S, Ono K, Rai G, Matsumura T. Contamination of soil with helminth parasite eggs in Nepal. Southeast Asian J Trop Med Public Health. 2000; 31(2):388–93. PMID: 11127345
- Uga S, Ono K, Kataoka N, Safriah A, Tantular IS, Dachlan YP, et al. Contamination of soil with parasite eggs in Surabaya, Indonesia. *Southeast Asian J Trop Med Public Health*. 1995; 26(4):730–4. PMID: 9139385
- Chongsuvivatwong V, Uga S, Nagnaen W. Soil contamination and infections by soil-transmitted helminths in an endemic village in southern Thailand. Southeast Asian J Trop Med Public Health. 1999; 30 (1):64–7. PMID: 10695791
- Ogbolu DO, Alli OA, Amoo AO, Olaosun II, Ilozavbie GW, Olusoga-Ogbolu FF. High-level parasitic contamination of soil sampled in Ibadan metropolis. *Afr J Med Med Sci.* 2011; 40(4):321–5. PMID: 22783681
- Muller M, Sánchez RM, Suswillo RR. Evaluation of a sanitation programme using eggs of Ascaris lumbricoides in household yard soils as indicators. J Trop Med Hyg. 1989; 92(1):10–6. PMID: 2918573
- Blaszkowska J, Kurnatowski P, Damiecka P. Contamination of the soil by eggs of geohelminths in rural areas of Lodz district (Poland). *Helminthology*. 2011; 48 (2):67–76. https://doi.org/10.2478/s11687-011-0012-8
- Horiuchi S, Paller VG, Uga S. Soil contamination by parasite eggs in rural village in the Philippines. *Trop Biomed.* 2013; 30(3):495–503. PMID: 24189679
- 34. Gyoten J, Hoa TVN, Fujimaki Y, Tanaka K, Uga S and Noda S. The correlation between contamination of soil with Ascaris sp. Eggs in school yards and ascariasis among primary school children in Mai Trung Commune Northen Veitnam. Trop Med Health. 2010; 38 (1).
- Vanhee M, Dalemans AC, Viaene J, Depuydt L, Claerebout E. *Toxocara* in sandpits of public playgrounds and kindergartens in Flanders (Belgium). *Vet Parasitol Reg Stud Reports*. 2015; 2: 51–54. https://doi.org/10.1016/j.vprsr.2016.03.002 PMID: 31018409
- Swetha CS, Supriya RA, Babu AJ, Sudhanthiramani T Rao TM. Prevalence of zoonotic important *Toxocara* eggs in soil samples of public places in and around Tirupati, Andhra Pradesh, India. The *Pharm Innov J.* 2017; 6(9): 330–333.
- Stojčević D, Sušić V, Lučinger S. Contamination of soil and sand with parasite elements as a risk factor for human health in public parks and playgrounds in Pula, Croatia. Vet archive. 2010; 80 (6), 733–742.
- **38.** Smith HV. Detection of parasites in the environment. *Parasitology*. 1998; 117(Suppl): S113–141. PMID: <u>10660936</u>
- Landa-Cansigno O, Durán-Álvarez JC, Jiménez-Cisneros B. Retention of *Escherichia coli*, *Giardia lamblia* cysts and *Ascaris lumbricoides* eggs in agricultural soils irrigated by untreated wastewater. *J Environ Manage*. 2013; 128:22–29. https://doi.org/10.1016/j.jenvman.2013.04.049 PMID: 23722173
- 40. Ajala MO, and Asaolu SO. Efficiency of the salt flotation technique in the recovery of Ascaris lumbricoides eggs from the soil. J Helminthol. 1995; 69:1–5. <u>https://doi.org/10.1017/s0022149x00013754</u> PMID: 7622785
- Nunes CM, Sinhorini IL, Ogassawara S. Influence of soil texture in the recovery of *Toxocara canis* eggs by a flotation method. *Vet Parasitol.* 1994; 53(3–4):269–74. https://doi.org/10.1016/0304-4017(94) 90190-2 PMID: 7975122
- 42. Nwoke EU, Ibiam GA, Odikamnoro OO, Umah OV, Ariom OT, Orji I. Examination of soil samples for the incidence of geohelminth parasites in Ebonyi north-central area of Ebonyi State, south-east of Nigeria. Arch Appl Sci Res. 2013; 5 (6):41–48
- Beaver PC. Observations on the epidemiology of ascariasis in a region of high hookworm endemicity. J Parasitol. 1952; 38:445–453. PMID: 12991137
- Zibaei M, Bahadory S, Cardillo N, Khatami AR. Soil contamination with eggs of *Toxocara* species in public parks of Karaj, Iran. *Int J Enteric Pathog*. 2017; 5(2), 45–48; <u>https://doi.org/10.15171/ijep.2017</u>.

- Pezeshki A, Haniloo A, Alejafar A, Mohammadi-Ghalehbin B. Detection of *Toxocara* spp. eggs in the soil of public places in and around of Ardabil city, Northwestern Iran. Iran *J Parasitol.* 2017; 12(1):136– 142. PMID: 28761471
- 46. Doi R, Itoh M, Chakhatrakan S, Uga S. Epidemiological investigation of parasitic infection of schoolchildren from six elementary schools in Sakon Nakhon province, Thailand. *Kobe J Med Sci.* 2017; 62(5): E120–E128. PMID: 28289269
- Moskvina TV, Bartkova AD, Ermolenko AV. Geohelminths eggs contamination of sandpits in Vladivostok, Russia. Asian Pac J Trop Med. 2016; 9(12): 1215–1217. https://doi.org/10.1016/j.apjtm.2016.11. 002 PMID: 27955750
- Ulukanligil M, Seyrek A, Aslan G, Ozbilge H, Atay S. Environmental pollution with soil-transmitted helminths in Sanliurfa, Turkey. *Mem Inst Oswaldo Cruz*. 2001; 96(7) Rio de https://doi.org/10.1590/s0074-02762001000700004 PMID: 11685253
- 49. Raissi V, Saber V, Zibaei M, Bahadory S, Akhlaghi E, Raiesi O, et al. Comparison of the prevalence of *Toxocara* spp. eggs in public parks soils in different seasons, from 2017 to 2018, Tehran province, Iran. *Clin Epidemiol Glob Health*. 2020; (8) 450–454.
- Motazedian H, Mehrabani D, Tabatabaee SH, Pakniat A, Tavalali M. Prevalence of helminth ova in soil samples from public places in Shiraz. *East Mediterr Health J*. 2006; 12(5):562–5. PMID: 17333794
- Azian MN, Sakhone L, Hakim SL, Yusri M, Nurulsyamzawaty Y, Zuhaizam A, et al. Detection of helminth infections in dogs and soil contaminations in rural and urban areas. *Southeast Asian J Trop Med Public Health.* 2008; 39 (2).
- Ghashaei O, Khedri J, Jahangeri-Nasr F, Hashemi SH. Contamination of soil samples of public parks with *Toxocara* spp. eggs in Kermanshah, Iran. J Fac Vet Med Istanbul Univ. 2016; 42 (1), 47–50.
- Gordon CA, Gray DJ, Gobert GN, McManus DP. DNA amplification approaches for the diagnosis of key parasitic helminth infections of humans. *Mol Cell Probes*. 2011; 25(4):143–52. <u>https://doi.org/10.1016/j.</u> mcp.2011.05.002 PMID: 21651977
- Ngui R, Lim YA, Traub R, Mahmud R, Mistam MS. Epidemiological and genetic data supporting the transmission of *Ancylostoma ceylanicum* among human and domestic animals. *PLoS Negl Trop Dis.* 2012; 6(2): e1522. https://doi.org/10.1371/journal.pntd.0001522 PMID: 22347515
- 55. George S, Levecke B, Kattula D, Velusamy V, Roy S, Geldhof P, et al. Molecular identification of hookworm isolates in humans, dogs and soil in a tribal area in Tamil Nadu, India. *PLoS Negl Trop Dis.* 2016; 10(8): e0004891. https://doi.org/10.1371/journal.pntd.0004891 PMID: 27486798
- 56. Bass D, Stentiford G, Littlewood DTJ, Hartikainen H. Diverse applications of environmental DNA methods in parasitology. *Trends Parasitol.* 2015; 31(10): 499–513. <u>https://doi.org/10.1016/j.pt.2015.06.013</u> PMID: 26433253
- 57. Gyawali P, Ahmed W, Sidhu JP, Nery SV, Clements AC, Traub R, et al. Quantitative detection of viable helminth ova from raw wastewater, human feces, and environmental soil samples using novel PMAqPCR methods. *Environ Sci Pollut Res* Int. 2016; 23(18):18639–48. https://doi.org/10.1007/s11356-016-7039-9 PMID: 27306209
- Collender PA, Kirby AE, Addiss DG, Freeman MC, Remais JV. Methods for quantification of soil-transmitted helminths in environmental media: current techniques and recent advances. *Trends Parasitol.* 2015; 31(12):625–639. https://doi.org/10.1016/j.pt.2015.08.007 PMID: 26440788
- 59. Mekonnen Z, Meka S, Ayana M, Bogers J, Vercruysse J, Levecke B. Comparison of individual and pooled stool samples for the assessment of soil-transmitted helminth infection intensity and drug efficacy. *PLoS Negl Trop Dis.* 2013; 7(5): e2189. https://doi.org/10.1371/journal.pntd.0002189 PMID: 23696905
- 60. Dana D, Vlaminck J, Ayana M, Tadege B, Mekonnen Z, Geldhof P, et al. Evaluation of copromicroscopy and serology to measure the exposure to *Ascaris* infections across age groups and to assess the impact of 3 years of biannual mass drug administration in Jimma Town, Ethiopia. *PLoS Negl Trop Dis.* 2020; 14(4); e0008037. https://doi.org/10.1371/journal.pntd.0008037 PMID: 32282815
- Ayana M, Cools P, Mekonnen Z, Biruksew A, Dana D, Rashwan N, et al. Comparison of four DNA extraction and three preservation protocols for the molecular detection and quantification of soil-transmitted helminths in stool. *PLoS Negl Trop Dis.* 2019; 13 (10): e0007778. https://doi.org/10.1371/ journal.pntd.0007778 PMID: 31658264
- 62. Cools P, Vlaminck J, Albonico M, Ame S, Ayana M, José Antonio BP, et al. Diagnostic performance of a single and duplicate Kato-Katz, Mini-FLOTAC, FECPAKG2 and qPCR for the detection and quantification of soil-transmitted helminths in three endemic countries. *PLoS Negl Trop Dis.* 2019; 13(8): e0007446. https://doi.org/10.1371/journal.pntd.0007446 PMID: 31369558
- Praet N, Verweij JJ, Mwape KE, Phiri IK, Muma JB, Zulu G, et al. Bayesian modelling to estimate the test characteristics of coprology, coproantigen ELISA and a novel real-time PCR for the diagnosis of taeniasis. *Trop Med Int Health*. 2013; 18(5):608–14. <u>https://doi.org/10.1111/tmi.12089</u> PMID: 23464616

- Levecke B, Montresor A, Albonico M, Ame SM, Behnke JM, Bethony JM, et al. Assessment of anthelmintic efficacy of mebendazole in school children in six countries where soil-transmitted helminths are endemic. *PLoS Negl Trop Dis.* 2014; 8(10): e3204. https://doi.org/10.1371/journal.pntd.0003204 PMID: 25299391
- Burden DJ, Whitehead A, Green EA, McFadzean JA, Beer RJ. The treatment of soil infested with the human whipworm, *Trichuris trichiura*. J Hyg. 1976; 77(3):377–82. https://doi.org/10.1017/ s0022172400055753 PMID: 1069817
- Katakura K, Hamada A, Kobayashi A. The fate of Ascaris eggs applied to the soil under various conditions. Jpn J Parasitol. 1986; 35(1) 1–9
- Liu J, Gratz J, Amour C, Kibiki G, Becker S, Janaki L, et al. A laboratory developed TaqMan Array Card for simultaneous detection of 19 enteropathogens. *J Clin Microbiol*. 2013; 51(2):472–80. https://doi.org/ 10.1128/JCM.02658-12 PMID: 23175269
- Alonzo TA, Pepe MS. Using a combination of reference tests to assess the accuracy of a new diagnostic test. *Stat Med* 1999; 18(22):2987–3003. https://doi.org/10.1002/(sici)1097-0258(19991130) 18:22<2987::aid-sim205>3.0.co;2-b PMID: 10544302
- Rosa Xavier IG, Ramos BC, Santarém VA. Recovery threshold of *Toxocara canis* eggs from soil. *Vet Parasitol* 2010; 167(1):77–80. https://doi.org/10.1016/j.vetpar.2009.09.052 PMID: 19879694
- 70. Quinn R, Smith HV, Bruce RG, Girdwood RW. Studies on the incidence of *Toxocara* and *Toxascaris* spp. ova in the environment. A comparison of flotation procedures for recovering *Toxocara* spp. ova from soil. *J Hyg.* 1980; 84:83–89. https://doi.org/10.1017/s0022172400026553 PMID: 7351480
- Centers for disease control and prevention. CDC 24/7 <u>https://www.cdc.gov/dpdx/trichuriasis/index.</u> html, December 19, 2017
- Jejaw A, Zeynudin A, Zemene E, Belay T. Status of intestinal parasitic infections among residents of Jimma Town, Ethiopia. *BMC Res Notes* 7, 502 (2014). <u>https://doi.org/10.1186/1756-0500-7-502</u> PMID: 25100301
- 73. Dermauw V, Dorny P, Braae UC, Devleesschauwer B, Robertson LJ, Saratsis A, et al. Epidemiology of *Taenia saginata* taeniosis/cysticercosis: a systematic review of the distribution in southern and eastern Africa. *Parasit Vectors*. 2018; 11 (578). https://doi.org/10.1186/s13071-018-3163-3 PMID: 30400948
- 74. Ng-Nguyen D, Stevenson MA, Dorny P, Gabriël S, Vo TV, Nguyen V-AT, et al. Comparison of a new multiplex real-time PCR with the Kato Katz thick smear and copro-antigen ELISA for the detection and differentiation of Taenia spp. in human stools. *PLoS Negl Trop Dis.* 2017; 11(7): e0005743. <u>https://doi.org/10.1371/journal.pntd.0005743</u> PMID: 28686662
- Patel NM, Tatar EL. Unusual colonoscopy finding: *Taenia saginata* proglottid. *World J Gastroenterol.* 2007; 13(41):5540–5541. https://doi.org/10.3748/wjg.v13.i41.5540 PMID: 17907306
- 76. Chukwuma MC, Ekejindu IM, Agbakoba NR, Ezeagwuna DA, Anaghalu IC, Nwosu DC. The Prevalence and risk factors of geohelminth infections among primary school children in Ebenebe Town, Anambra State, Nigeria. *Middle East J Sci Res.* 2009; 4 (3): 211–215, ISSN 1990-9233
- 77. Odoba MB, Otalu OJ, Balogun JB. Prevalence of helminth parasite eggs in pupils and playing grounds of some selected primary schools in Zaria, Nigeria. *WJLSMR*. 2012; 2(5):192
- 78. Joël ATR, Jeannette Y, Gabriel TH, Arlette NT, Rosine NV, Mbida M. Rate of soil contamination by soil transmitted helminths in relation to the awareness status of pupils in primary schools of Penka-Michel, West-Cameroon. JHER. 2021; (72), 2021; 112–121. https://doi.org/10.11648/j.jher.20210702.15
- 79. Debalke S, Worku A, Jahur N, Mekonnen Z. Soil transmitted helminths and associated factors among school children in government and private primary school in Jimma Town Southwest Ethiopia. *Ethiop J Health Sci.* 2013; 23 (3) https://doi.org/10.4314/ejhs.v23i3.6 PMID: 24307823
- Sumo L, Otiobo Atibita EN, Mache E, Gangue T, Nana-Djeunga HC. Transmission of soil transmitted helminthiasis in the Mifi health district (West Region, Cameroon): low endemicity but still prevailing risk. *Parasitology*. 2021, 1:95–104. https://doi.org/10.3390/parasitologia1030011
- Regulations E. Technology: Control of pathogens and vector attraction in sewage sludge. USEPA, Office of Research and Development. 2003.
- World Health Organization. Guidelines for the safe use of wastewater, excreta and greywater in agriculture and aquaculture. World Health Organization 2006. https://apps.who.int/iris/handle/10665/78265